Author’s response to reviews

Title: Consent to Data Linkage in a Large Online Epidemiological Survey of 18-23 Year Old Australian Women in 2012-13

Authors:

Deirdre Mclauglin (deirdre.mclauglin@unimelb.edu.au)

Anna Graves (anna.graves@newcastle.edu.au)

Janni Leung (j.leung1@uq.edu.au)

Jennifer Powers (jenny.powers@newcastle.edu.au)

Version: 4 Date: 24 Nov 2019

Author’s response to reviews:

1. Authorship Change

We note that the order of authors has changed since original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.

The Change of Authorship Request Form has been completed and uploaded as supplementary material. It has also been emailed on two previous occasions to the Editorial Office.

2. Authors’ contributions - author approval

Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript in its current state.

This has been done.
3. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

This has been done.